Author's response to reviews

Title: Antibiotic prophylaxis for caesarean section at a Ugandan Hospital: a randomised clinical trial evaluating the effect of administration time on the incidence of post-operative infections

Authors:

Lomangisi D Dlamini (ngisi2@yahoo.com)
Musa Sekikubo (msekikubo@yahoo.com)
Janat Tumukunde (janattumukunde@gmail.com)
Charles Kojjo (charleskojjo@gmail.com)
Davidson Ocen (d4davidson@gmail.com)
Arthur Kwizera (akwizera@chs.mak.ac.ug)

Version: 4 Date: 10 January 2015

Author's response to reviews: see over
Response to Editorial comments

Title: Antibiotic prophylaxis for caesarean section at a Ugandan hospital: a randomised clinical trial evaluating the effect of administration time on the incidence of post-operative infections

Dear Editorial Team

Thank you for assessing our paper. It was recommended that we copyedit our paper or use a professional editing service.

We submitted our paper for editing to Edanz. The changes that were made are as follows:

- Page 10 line 13 sentence changed by including ‘statistically’ significant
- Page 10 lines 16 – 17 was edited to make more sense
- Page 10 line 20 “the study” was inserted.

The manuscript was also checked if it was still conforming to journal guidelines.